Author's response to reviews

Title: Peripheral neurolymphomatosis with tracheal asphyxia: a case report and literature review

Authors:

zuofeng liu (zfeng_liu@sina.com)
tao jiang (suyeku@126.com)
ni hou (hou_n@sina.com)
yongqian jia (jia_yq@163.com)

Version: 4 Date: 12 June 2015

Author's response to reviews:

1. About our diagnosis:
   Though our patient refused to have a nerve biopsy, considering her lymphoma disease history, the symptoms of neurologic impairment and the PET/CT results, the diagnosis could be set up. It has also been reported that NL could be diagnosed without nerve biopsy.

2. About the past medical history:
   I gave a detailed description about the patient’ past medical history just because I guess that Our case may be suffered the brachial plexus NL at first administration for her right-sided brachial plexus nerve damage symptoms and with neurogenic damage showed by EMG.

3. About the complete remission:
   In our case, after the chemotherapy and radiotherapy the patient’ symptoms and sign resolved, PET/CT showed complete remission, so we think our patient get the complete remission.